Author's response to reviews

Title: Predictive Properties of the A-TAC Inventory When Screening for Childhood-onset Neurodevelopmental Problems in a Population-based Sample

Authors:

Tomas Larson (tomas.larson@med.lu.se)
Sebastian Lundström (sebastian.lundstrom@neuro.gu.se)
Thomas Nilsson (thomas.nilsson@neuro.gu.se)
Eva Norén Selinus (eva.noren-selinus@sll.se)
Maria Råstam (maria.rastam@med.lu.se)
Paul Lichtenstein (paul.lichtenstein@ki.se)
Clara Hellner Gumpert (clara.gumpert@ki.se)
Henrik Anckarsäter (henrik.anckarsater@neuro.gu.se)
Nóra Kerekes (nora.kerekes@neuro.gu.se)

Version: 2 Date: 18 October 2012

Author's response to reviews:

Dear Dr Olino

Thank you for pointing out the formatting errors in the manuscript. Two sentences have now been added as complements to the "Ethical Considerations" paragraph.

1. Under the heading "Subjects at Baseline"
"All subjects in CATSS are covered by the informed consent procedure, and a telephone interview is conducted with the parents of all participating twins around the children’s 9th or 12th birthdays”.

2. Under the heading "Subjects at Follow-up"
"A written informed consent for participation in CATSS-15/DOGSS was obtained from all subjects, parents as well as adolescents”.

Yours sincerely,
Tomas Larson